

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2021GL095494

Key Points:

- The amplitude of the annual temperature cycle decreased over the Tibetan Plateau, and this diminished trend was amplified with elevation
- Influence of anthropogenic forcing on the elevation-dependent weakening of annual temperature cycle amplitude is detectable
- The increase in anthropogenic aerosols may be the main contributor

Correspondence to:

L. Zhu, zhulh@cuit.edu.cn

Citation:

Zhu, L., Huang, G., Fan, G., & Hua, W. (2022). Influence of anthropogenic activities on elevation-dependent weakening of annual temperature cycle amplitude over the Tibetan Plateau. *Geophysical Research Letters*, *49*, e2021GL095494. https://doi. org/10.1029/2021GL095494

Received 11 AUG 2021 Accepted 31 MAR 2022

Author Contributions:

Conceptualization: Lihua Zhu Data curation: Lihua Zhu Formal analysis: Lihua Zhu Funding acquisition: Guangzhou Fan Investigation: Lihua Zhu Methodology: Lihua Zhu Project Administration: Guangzhou Fan Resources: Wei Hua Software: Lihua Zhu Supervision: Gang Huang Validation: Lihua Zhu Visualization: Lihua Zhu Writing – original draft: Lihua Zhu

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Influence of Anthropogenic Activities on Elevation-Dependent Weakening of Annual Temperature Cycle Amplitude Over the Tibetan Plateau

Lihua Zhu^{1,2}, Gang Huang^{3,4,5,6}, Guangzhou Fan¹, and Wei Hua¹

¹School of Atmospheric Sciences/Plateau Atmosphere and Environment Key Laboratory of Sichuan Province/Joint Laboratory of Climate and Environment Change, Chengdu University of Information Technology, Chengdu, China, ²Heavy Rain and Drought-Flood Disasters in Plateau and Basin Key Laboratory of Sichuan Province, Chengdu, China, ³State Key Laboratory of Numerical Modeling for Atmospheric Sciences and Geophysical Fluid Dynamics, Institute of Atmospheric Physics, Chinese Academy of Sciences, Beijing, China, ⁴Laboratory for Regional Oceanography and Numerical Modeling, Qingdao National Laboratory for Marine Science and Technology, Qingda, China, ⁵Joint Center for Global Change Studies (JCGCS), Beijing, China, ⁶University of Chinese Academy of Sciences, Beijing, China

Abstract The annual cycle is the dominant component and the most prominent climate oscillation for the temperature variation over the Tibetan Plateau (TP). Identifying the relative contribution of anthropogenic and natural forcing to the changes in the annual cycle, namely, detection and attribution, is an important aspect for mountainous climate change research. The present study documents the elevation dependence of the trend in the annual temperature cycle (ATC) amplitude over the TP and detects the influence of anthropogenic activities on it. An elevation-dependent weakening of the ATC amplitude occurs over the TP during the period 1961–2014. This variation with altitude can mainly be ascribed to the seasonal difference in warming. The influence of anthropogenic activities is detectable, with increased aerosols being the main contributor. Under aerosol-only forcing, the larger decrease in snow-related albedo at higher altitudes in winter can explain the amplified negative tendency of the ATC amplitude with elevation.

Plain Language Summary The present study documents the variation with altitude of the trend in the seasonal temperature difference over the TP and detects the influence upon it of anthropogenic activities. The seasonal temperature difference weakened in most regions of the TP during 1961–2014. Also, the higher the altitude, the more notable the decrease. The trend in the temperature seasonality over the TP and its variation with altitude can mainly be ascribed to the difference in the rate of warming between winter and summer. The greater warming in winter in higher-altitude regions over the TP causes the negative tendency of the seasonal temperature difference to amplify with elevation. The model-simulated responses are able to capture the observed variation with altitude of the trend in the temperature seasonality only if anthropogenic forcing is involved. Moreover, the influence upon it of anthropogenic activities is statistically detectable, with the increase in anthropogenic aerosols being the main contributor. In the model-simulated response to anthropogenic aerosol–only forcing, the larger decrease in snow-related albedo at higher altitudes in winter can explain the amplified warming there in winter and thereby the weakening with elevation in the seasonal temperature difference.

1. Introduction

The climatic responses to global warming in mountainous regions and their elevation dependence are hot topics internationally among scientists working in mountain climate research (Pepin et al., 2015; Rangwala & Miller, 2012; You et al., 2020). Mountain areas act as amplifiers of global warming, and an increasing number of studies have shown that climate warming is enhanced with altitude (e.g., Giorgi et al., 1997; Minder et al., 2018; Pepin et al., 2015; Rangwala et al., 2010). Among the numerous mountainous regions of the world, the Tibetan Plateau (TP) is well known for its vast area, having some of the highest altitudes in the world, and extremely complex terrain. The response of the TP to climate change is more sensitive than that of other regions at the same latitude (Qiu, 2008; Yao, Thompson, Mosbrugger, et al., 2012, Yao, Thompson, Yang, et al., 2012) and it is closely related to altitude (e.g., Palazzi et al., 2017; Pepin et al., 2015; Rangwala et al., 2009, 2013; Thakuri et al., 2019), especially in winter. Compared with other seasons, both the highest rate of warming and the most significant elevation-dependent warming over the TP occur in winter (Dimri et al., 2018; Liu & Chen, 2000;

Liu et al., 2009). Accordingly, an interesting question arises. Has the amplitude of the annual temperature cycle (ATC) over the TP decreased owing to the difference in the rate of warming between winter and summer? If the amplitude has decreased, does this weakening trend in temperature seasonality vary with altitude as a result of the seasonal difference in elevation-dependent warming?

The annual cycle refers to the dominant variability of numerous climate variables in the extratropics at the daily and monthly time scales (Qian, Fu, et al., 2011; Qian, Wu, et al., 2011; Qian & Zhang, 2015; Stine et al., 2009; Stine & Huybers, 2012; Thomson, 1995). A growing number of studies have shown that climate change is reflected not only in the change of the annual mean for a climate variable but also in the long-term change in the amplitude and phase of its annual cycle (e.g., Stine et al., 2009; Thomson, 1995, 2009; Wang & Dillon, 2014). Changes in the amplitude of the ATC influence the estimation of climate variability and trends (Qian, Fu, et al., 2011; Thompson, 1995), such as the classification of El Niño/La Niña years (Qian, Wu, et al., 2011). Taking into account changes in the amplitude of the ATC in temperature reconstructions have also been suggested (Jones et al., 2003). Moreover, changes in the ATC could strongly affect biological and ecological systems, such as bird and mammal distributions (Porter et al., 2000), plant developmental genetics (Li et al., 2010), and insect population dynamics (Vasseur et al., 2014).

The TP is species-rich, and the distribution of animals and plants has obvious features of transition with altitude. Climate change in the different altitudinal zones within a mountain range has important implications for ecosystems, farming and species along an elevation gradient (e.g., Freeman et al., 2018; González-Orozco & Porcel, 2021; Liancourt et al., 2020; Wen et al., 2018), as well as the high-elevation cryosphere and associated runoff and even the source of water for large populations in lower-elevation regions (e.g., Immerzeel et al., 2010, 2012; Yao, Thompson, Mosbrugger, et al., 2012; Yao, Thompson, Yang, et al., 2012; Yao, et al., 2019). The social, economic and eco-environmental impacts arising from altitude-related climate change over the TP could therefore be large, hence justifying further research. Elevation-dependent warming in mean temperature and its mechanisms (e.g., Minder et al., 2018; Pepin et al., 2015; You et al., 2020) have been well documented. However, knowledge of mean temperature alone has important limitations when applied to understanding the effects of environmental variance on ecological and evolutionary processes (Ruel & Ayres, 1999). Changes in temperature variation can have profound effects that match or even exceed the impacts of mean temperature (e.g., Bauerfeind & Fischer, 2014; Paaijmans et al., 2010, 2013; Seddon et al., 2016; Vasseur et al., 2014). For instance, temperature seasonality and topography are more important indicators of species diversity than mean annual temperature (Shrestha et al., 2018). Variables related to temperature seasonality seem to be more important than variables related to mean temperature for distinguishing biomes (Silva de Miranda et al., 2018). Compared to the mean annual temperature, frost, which is correlated with temperature seasonality (Hänninen, 2016), is identified as a more important driver of tree growth (Marquis et al., 2020), and seems to better explain the altitudinal and latitudinal range limits of tree species (Du et al., 2019; Kollas et al., 2014; Körner et al., 2016; Vitra et al., 2017). Therefore, knowledge of not only changes in mean temperatures but also changes in temperature variation is required to reasonably understand and predict the biological consequences of climate change (Karl et al., 1995).

In recent decades, as the dominant component and the most prominent climate oscillation, the amplitude of the ATC has displayed a decreasing trend in much of the Northern Hemisphere middle-to high-latitude land areas (e.g., Qian & Zhang, 2015; Stine et al., 2009; Thomson, 1995; Wang & Dillon, 2014). Furthermore, anthropogenic forcing has been indicated to impose a detectable influence on the changes in the ATC in certain regions (Mann & Park, 1996; Qian & Zhang, 2015; Santer et al., 2018; Wallace & Osborn, 2002). With regard specifically to the ATC of the TP, many uncertainties and unanswered questions remain. For instance, although a reduced ATC amplitude has been observed over the TP above 4,000 m (Duan et al., 2017), it remains ambiguous how the ATC amplitude at other altitudes of the TP has changed, whether this variation is related to elevation, and whether human activity has a detectable influence. These questions will be addressed in this paper by analyzing and comparing observed and model-simulated changes with altitude in the amplitude trend of the ATC over the TP.

2. Data and Methods

The regular surface meteorological observations for surface air temperature (SAT) applied here are provided by the China Meteorological Administration. The present analysis is based on these data over the TP above 1,000 m with 101 stations for the period of 1961–2014. It comprises the same data records as the model output discussed

List of CMIP6 Models, Experiments, and Ensemble Numbers for This Study							
No.	Models	Historical	Hist-nat	Hist-GHG	Hist-aer (SAT/RSDS/RSUS)	Hist-aer SNC	piControl
1	ACCESS-ESM1-5	7	3	1	1	0	18
2	CESM2	9	3	3	2	2	22
3	CNRM-CM6-1	4	4	4	4	4	9
4	FGOALS-g3	1	1	1	1	0	13
5	GFDL-ESM4	1	1	1	1	1	9
6	GISS-E2-1-G	5	5	4	4	4	30
7	HadGEM3-GC31-LL	4	4	4	4	4	9
8	IPSL-CM6A-LR	6	6	2	2	2	41
9	MIROC6	10	10	1	4	4	15
10	MRI-ESM2-0	7	5	5	5	5	17
11	NorESM2-LM	3	3	3	3	3	8
	Sum (models)	57 (11)	45 (11)	29 (11)	31 (11)	29 (9)	191 (11)

 Table 1

 List of CMIP6 Models, Experiments, and Ensemble Numbers for This Study.

Note. Figures in parentheses in the last row indicate the total number of models for each forcing.

in the following study. Missing values account for less than 0.5% of the total records during the research period; hence, the quality of the data is reasonable. Moreover, missing values in this study were addressed using the method applied in Duan and Wu (2008).

Monthly mean SAT from models participating in the World Climate Research Program's Coupled Model Intercomparison Project Phase 6 (CMIP6; Eyring et al., 2016; https://esgf-node.llnl.gov/projects/cmip6/) are applied in this study. Among them, historical simulations from 1961 to 2014 are employed to estimate the responses of model-simulated temperature to different external forcings. Preindustrial control simulations (piControl), which are divided into multiples of nonoverlapping 54-year intervals, are adopted to assess natural internal climate variability. The climate model-simulated responses to external forcings considered here are obtained from five different experiments as follows: all-forcing simulation (ALL), historical natural-only run (NAT), anthropogenic forcings simulation (ANT), historical well-mixed greenhouse gases forcing only (GHG), and historical anthropogenic aerosol-only run (AER). The model-simulated responses to anthropogenic forcings are obtained by subtracting the natural forcing from the all-forcing simulation (e.g., Hu & Sun, 2021; Paik & Min, 2020; Seong et al., 2021). Eleven models are selected as they provide both historical simulations under all abovementioned and preindustrial control simulations, as well as surface altitude (Table 1). Moreover, due to the lack of surface albedo data in simulation output, monthly mean surface downwelling shortwave radiation (RSDS) and surface upwelling shortwave radiation (RSUS) under anthropogenic aerosol-only forcing are also employed in our study to calculate surface albedo, which is expressed as the ratio between the RSUS and the RSDS. To ensure equal weights for different models, the multimodel ensemble mean is obtained by calculating the individual model ensemble mean and then averaging across multiple models.

To quantify the possible influence of external forcings on the elevation dependence of the amplitude trend, we use a correlation-based method of detection analyses here, which is applied in many studies (e.g., Qian & Zhang, 2015; Santer et al., 1995; Wan et al., 2015). The series for mean amplitude trend values on non-overlapping altitudinal ranges in observations and model-simulated responses to the abovementioned external forcings are compared by computing correlation coefficients between observations and simulations. To determine the statistical significance of the correlations, we also calculate the correlation coefficients between observations and the model-simulated response to a forcing is above the 90th percentile of the correlation coefficients between the observations and the piControl intervals, that correlation is considered to be statistically significant at the 90% confidence level over internal variability, and the forced response is detectable in the observations. We need to address a large number of model simulations in this study. The seasonal maximum for SAT over the TP occurs

in JJA, and the minimum occurs in DJF. Hence, the amplitude of the ATC is calculated via a simplified scheme mentioned in the literature (Qian & Zhang, 2015):

$$A = \frac{(T_{\rm JJA} - T_{\rm DJF})}{2} \tag{1}$$

where A is the amplitude, and T_{JJA} and T_{DJF} are the temperature in the current JJA (June–July–August) and the temperature in the following DJF (December–January–February), respectively.

3. Results

3.1. Elevation-Dependent Weakening of the ATC Amplitude

The amplitude of the ATC for most stations over the TP shows a decreasing trend during the 1961/1962–2013/2014 period (Figure 1a). More importantly, larger weakening trends tend to occur in higher-altitude regions, especially above 2,500 m, where the trends averaged within every altitudinal range are all significant at the 95% confidence level (Figure 1b). That is, an elevation-dependent weakening of the ATC amplitude is apparent over the TP. This significant elevation dependence for the amplitude trend is clearly demonstrated in Figure 1c, where the correlation coefficient between the trend of the ATC amplitude and the elevation for the 101 stations is -0.24 (p < 0.05).

To investigate the cause of the changes with altitude in the ATC amplitude, the relationships between the trend of the ATC amplitude and the annual or seasonal temperature trends over the TP are analyzed in Figures 2a–2c. The trend of the ATC amplitude over the TP shows a significant correlation with the annual mean temperature trend (Figure 2a) and the winter temperature trend (Figure 2c), with the correlation coefficients of -0.444(p < 0.01) and -0.739 (p < 0.01), respectively. However, the correlation between the summer temperature trend and the amplitude trend is not significant (Figure 2b). In general, the SAT for annual mean, summer and winter presents an increasing trend over the TP (Figures 2d–2f), and the warming in winter is more striking than that in summer, which contributes to the diminishing seasonal temperature difference in Figure 1. More notably, the SAT over the TP represents elevation-dependent warming in winter (Figure 2f), and the correlation coefficient between the winter temperature trend and the altitude is 0.311 (p < 0.1). For summer, the result is completely different (Figure 2e). The correlation between elevation and the summer temperature trend is not significant. Thus, the greater warming in winter in higher-altitude regions over the TP causes the negative tendency of the ATC amplitude to amplify with elevation (Figure 1).

3.2. Detection and Attribution

Detection and attribution analyses are performed for the central-eastern TP (CE-TP, $25^{\circ}N-40^{\circ}N$, $85^{\circ}E-106^{\circ}E$) as most of the meteorological stations are located there. Amplitude trend anomalies in Figure 3 are calculated by removing the regional mean values over the CE-TP above 1,000 m. The altitudinal range is further refined to obtain more samples. The multimodel ensemble mean under ALL forcing can capture the observed change in the trend of the ATC amplitude with altitude, and similar detection results are obtained when the ANT forcing signal is estimated (Figure 3a). The responses to NAT forcing do not show these characteristics. The results suggest that the elevation-dependent variation in the amplitude trend is consistent with the observations only if anthropogenic forcing is involved.

Moreover, the model-simulated response to ALL or ANT forcing is detectable in Figure 3b, whereas that to NAT forcing is not detectable. The correlation between the observations and the model-simulated responses, including to ALL forcing and to ANT forcing, is significant, as tested by a one-sided Monte Carlo test conducted with piControl simulations (Figure 3b), with correlation coefficients of 0.771 and 0.756, respectively. This finding indicates that the responses to ALL and ANT forcing can be detected at the 95% confidence level, whereas the NAT signal cannot be detected. Furthermore, the trend of the ATC amplitude in the AER response displays a discernible elevation-dependent weakening which is similar to that observed (Figure 3c), with a correlation coefficient of 0.732 (p < 0.05), while the correlation between the observations and the response to GHG forcing is not significant (Figure 3c), with a correlation coefficient of 0.627 (p > 0.1). It appears that the detectable variation in the ANT response is primarily derived from AER forcing.



1 km <= H < 1.5 km
1.5 km <= H < 2 km
2 km <= H < 2.5 km
2.5 km <= H < 3 km
3 km <= H < 3.5 km
3.5 km <= H < 4 km
4 km <= H < 4.5 km
4.5 km <= H < 5 km



Figure 1. (a) Spatial distribution of the linear trend of the annual temperature cycle (ATC) amplitude over the Tibetan Plateau (TP) at altitudes above 1,000 m. The open circles represent negative values, and the solid circles represent positive values. The larger is the circle, the larger the absolute value. The black curve outlines the TP area above 2,000 m. (b) Average trend of the ATC amplitude for every altitudinal range (line). Values exceeding the 95% confidence level are presented in dots. Bars represent mean altitudes. (c) Relationships between the trend of the ATC amplitude and the altitude over the TP. The solid circles represent the stations where ATC amplitude trends exceed the 90% significance level.

In general, the similarity in the changes between the observations and the model-simulated responses indicates that the influence of anthropogenic activities on the elevation-dependent weakening of the ATC amplitude is detectable and that the increase in AER is the main contributor.

The TP is located in the immediate vicinity of densely populated and industrialized regions. Growing evidence has demonstrated that exogenous air pollutants can enter the TP's environments and impact the cryosphere (e.g., Di Mauro, 2020; Kang et al., 2019; Li et al., 2016; Sarangi et al., 2020; Xu et al., 2009). Carbonaceous particles and dust can be transported from the west and southwest to the hinterlands of the TP by prevailing wind (Ji et al., 2015; Kang et al., 2019; Xia et al., 2011). Reduced Arctic sea ice intensifies aerosol transport to the TP (Li et al., 2020). The deposition of these light-absorbing particles can substantially accelerate snowmelt (Gautam



Figure 2. Relationships between the trend of the annual temperature cycle (ATC) amplitude and the (a) annual temperature trend, (b) summer temperature trend, and (c) winter temperature trend, as well as relationships between the (d) annual temperature trend, (e) summer temperature trend, and (f) winter temperature trend and the altitude over the Tibetan Plateau (TP).

et al., 2013; Lau et al., 2010; McKenzie Skiles & Painter, 2018; Painter et al., 2013; Qian et al., 2015; Xu et al., 2009). Moreover, dust aerosols in the atmosphere warm clouds and increase the evaporation of cloud droplets, eventually leading to less precipitation (Huang et al., 2014). Enhanced snow loss at higher elevations over the TP cause development of elevation-dependent warming, especially in winter (Guo et al., 2021). Accordingly, trends in snow area percentage (SNC), albedo and SAT over the TP for model-simulated responses to AER forcing and their relationships were analyzed to demonstrate the impact of aerosols on ATC amplitude. The SNC over the TP in the model-simulated responses to AER forcing decreased strikingly as the altitude increased in winter, particularly above 2,500 m (Figure 4a). This finding may be relevant to the signature of the increasing dust-induced snow darkening with surface elevation over High-Mountain Asia (Sarangi et al., 2020), which enhances snowmelt trends over high-altitude regions. The slight trend of SNC in summer may be related to the seasonal variation in aerosol deposition (Chen et al., 2015; Cong et al., 2015; Kang et al., 2019; Wan et al., 2017). Seasonal variation in altitude-dependent SNC trends could cause the albedo of the high elevation decrease in winter to be more obvious than that in summer (Figure 4b), followed by a signature of amplified warming with elevation in winter (Figure 4c). The correlation between the albedo trend and the SAT trend is significant in both summer and winter (Figures 4d and 4e), with correlation coefficients of -0.45 (p < 0.1) and -0.686 (p < 0.01), respectively. This finding confirms the conclusion that aerosols may have made an important contribution to elevation-dependent warming over the TP (Kang et al., 2019; Xu et al., 2009; You et al., 2020). The greater warming in winter in higher-altitude regions acts as the main contributor to the elevation-dependent negative tendency of the ATC





Figure 3. (a) The 200-m altitudinal-range mean trend anomalies of the annual temperature cycle (ATC) amplitude in the observations and model-simulated responses to all-forcing simulation (ALL), historical natural-only run (NAT), and anthropogenic forcings simulation (ANT) forcing are shown. The range of the 5th–95th percentile amplitudes simulated from 57 individual model runs under ALL forcing are shown in gray. (b) Correlation coefficients between the observed average trend and model-simulated average trend of the ATC amplitude for the 19 altitudinal ranges. The red triangle, blue spot, and orange diamond mark the correlation coefficients between the observations and model-simulated responses to ALL forcing, ANT forcing and NAT forcing, respectively. The 5th–95th and 10th–90th percentile ranges of the correlation coefficients between the observations and 191 intervals of piControl simulations are expressed by black error bars and green error bars, respectively. (c) The 200-m altitudinal-range mean trend anomalies of the ATC amplitude in the model-simulated responses to anthropogenic aerosol-only run (AER) and greenhouse gases forcing only (GHG) forcing are shown. Amplitude trend anomalies are calculated by removing the regional mean values over the CE-TP above 1,000 m.

amplitude (Figures 2c and 2f). Quantitatively, compared with summer (Figure 4f), the decreasing albedo over high regions in winter, as a consequence of elevation-dependent shrinkage in SNC, is the main contributor to the amplified negative tendency with elevation for the ATC amplitude in the model-simulated response to AER forcing (Figure 4g). The correlation between the ATC amplitude trend and the winter albedo trend is significant at the 99% confidence level, with a correlation coefficient of 0.763 (Figure 4g). However, the correlation between the amplitude trend and the summer albedo trend was not statistically significant (Figure 4f).





Figure 4. The 200-m altitudinal-range mean (a) snow area percentage (SNC) trend, (b) albedo trend and (c) SAT trend anomaly in the model-simulated responses to anthropogenic aerosol-only run (AER) forcing (d and e) Relationship between the SAT trend and albedo trend in (d) summer and (e) winter over the CE-TP at altitudes above 1,000 m for the 1961/1962–2013/2014 period (f and g) As shown in (d and e) but for the relationship between the trend of the annual temperature cycle (ATC) amplitude and the albedo trend in (f) summer and (g) winter.

4. Conclusions

The amplitude of the ATC decreased in most areas of the TP during the period 1961–2014; this diminished trend was amplified with elevation. Compared with the trend in the annual mean temperature and summer temperature, the warming in winter and its elevation dependence may to a large extent account for the negative tendency in the ATC amplitude and its amplification with altitude. The influence of anthropogenic forcing on the elevation-dependent weakening in the ATC amplitude over the TP can be separated from that of natural forcing, and the increase in anthropogenic aerosols may be the main contributor. In the model-simulated response to anthropogenic aerosol–only forcing, aerosols force a greater reduction in SNC and even surface albedo at higher altitudes in winter and thus enhance elevation-dependent warming in that season, thereby resulting in the negative



tendency of the ATC amplitude to enlarge with altitude over the TP. Additionally, further studies on the impact of aerosols on snowmelt and snowfall, as well as their relative contribution to shrinking SNC at high elevations, would be conducted by combining application of satellite-based datasets and modeling experiments to understand the complex physical mechanism of cryospheric changes associated with aerosols.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The CMIP6 models used in this study are listed in Table 1, and the model outputs are available at https://esgfnode.llnl.gov/search/cmip6/. The calculated trends of ATC amplitude based on original meteorological station data are available at https://doi.org/10.7910/DVN/87X4VY.

References

- Bauerfeind, S. S., & Fischer, K. (2014). Simulating climate change: Temperature extremes but not means diminish performance in a widespread butterfly. *Population Ecology*, 56(1), 239–250. https://doi.org/10.1007/s10144-013-0409-y
- Chen, P. F., Kang, S. C., Li, C. L., Rupakheti, M., Yan, F., Li, Q., et al. (2015). Characteristics and sources of polycyclic aromatic hydrocarbons in atmospheric aerosols in the Kathmandu Valley, Nepal. *The Science of the Total Environment*, 538, 86–92. https://doi.org/10.1016/j. scitotenv.2015.08.006
- Cong, Z., Kang, S., Kawamura, K., Liu, B., Wan, X., Wang, Z., et al. (2015). Carbonaceous aerosols on the south edge of the Tibetan plateau: Concentrations, seasonality and sources. *Atmospheric Chemistry and Physics*, *15*(3), 1573–1584. https://doi.org/10.5194/acp-15-1573-2015
- Di Mauro, B. (2020). A darker cryosphere in a warming world. Nature Climate Change, 10, 979–980. https://doi.org/10.1038/s41558-020-00911-9
 Dimri, A. P., Kumar, D., Choudhary, A., & Maharana, P. (2018). Future changes over the Himalayas: Mean temperature. Global and Planetary Change, 162, 235–251. https://doi.org/10.1016/j.gloplacha.2018.01.014
- Du, J., Li, K., He, Z., Chen, L., Zhu, X., & Lin, P. (2019). Age-mediation of tree-growth responses to experimental warming in the northeastern Tibetan Plateau. *Ecology and Evolution*, 9, 2242–2254. https://doi.org/10.1002/ece3.4920
- Duan, A. M., & Wu, G. X. (2008). Weakening trend in the atmospheric heat source over the Tibetan plateau during recent decades. Part I: Observations. Journal of Climate, 21(13), 3149–3164. https://doi.org/10.1175/2007JCL11912.1
- Duan, J. P., Esper, J., Büntgen, U., Li, L., Xoplaki, E., Zhang, H., et al. (2017). Weakening of annual temperature cycle over the Tibetan Plateau since the 1870s. *Nature Communications*, 8, 1–7. https://doi.org/10.1038/ncomms14008
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the coupled model Intercomparison Project phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9, 1937–1958. https://doi.org/10.5194/ GMD-9-1937-2016
- Freeman, B. G., Lee-Yaw, J. A., Sunday, J. M., & Hargreaves, A. L. (2018). Expanding, shifting and shrinking: The impact of global warming on species' elevational distributions. *Global Ecology and Biogeography*, 27(11), 1268–1276. https://doi.org/10.1111/geb.12774

Gautam, R., Hsu, N. C., Lau, W. K., & Yasunari, T. (2013). Satellite observations of desert dust-induced Himalayan snow darkening. Journal of Geophysical Research, 40(5), 988–993. https://doi.org/10.1002/grl.50226

- Giorgi, F., Hurrell, J., Marinucci, M., & Beniston, M. (1997). Elevation dependency of the surface climate change signal: A model study. Journal of Climate, 10(2), 288–296. https://doi.org/10.1175/1520-0442(1997)010<0288:edotsc>2.0.co;2
- González-Orozco, C. E., Porcel, M., & Palmeirim, A. F. (2021). Two centuries of changes in Andean crop distribution. Journal of Biogeography, 48, 1972–1980. https://doi.org/10.1111/jbi.14126
- Guo, D., Pepin, N., Yang, K., Sun, J., & Li, D. (2021). Local changes in snow depth dominate the evolving pattern of elevation-dependent warming on the Tibetan Plateau. Science Bulletin, 66(11), 1146–1150. https://doi.org/10.1016/j.scib.2021.02.013
- Hänninen, H. (2016). Boreal and temperature trees in a changing climate, modelling the ecophysiology of seasonality. Springer.
- Hu, T., & Sun, Y. (2021). Anthropogenic influence on extreme temperatures in China based on CMIP6 models. International Journal of Climatology, 1, 2981–2995, 1–15. https://doi.org/10.1002/joc.7402
- Huang, J., Wang, T., Wang, W., Li, Z., & Yan, H. (2014). Climate effects of dust aerosols over East Asian arid and semiarid regions. Journal of Geophysical Research: Atmospheres, 119(19), 11398–11416. https://doi.org/10.1002/2014JD021796
- Immerzeel, W. W., Van Beek, L. P. H., & Bierkens, M. F. P. (2010). Climate change will affect the Asian water towers. *Science*, 328(5984), 1382–1385. https://doi.org/10.1126/science.1183188
- Immerzeel, W. W., Van Beek, L. P. H., Konz, M., Shrestha, A. B., & Bierkens, M. F. P. (2012). Hydrological response to climate change in a glacierized catchment in the Himalayas. *Climatic Change*, 110(3), 721–736. https://doi.org/10.1007/s10584-011-0143-4
- Ji, Z., Kang, S., Cong, Z., Zhang, Q., & Yao, T. (2015). Simulation of carbonaceous aerosols over the third Pole and adjacent regions: Distribution, transportation, deposition, and climatic effects. *Climate Dynamics*, 45(9–10), 2831–2846. https://doi.org/10.1007/s00382-015-2509-1
- Jones, P. D., Briffa, K. R., & Osborn, T. J. (2003). Changes in the Northern Hemisphere annual cycle: Implications for paleoclimatology? Journal of Geophysical Research, 108(D18), 4588. https://doi.org/10.1029/2003JD003695
- Kang, S. C., Zhang, Q. G., Qian, Y., Ji, Z., Li, C., Cong, Z., et al. (2019). Linking atmospheric pollution to cryospheric change in the third Pole region: Current progress and future prospects. *National Science Review*, 6, 796–809. https://doi.org/10.1093/nsr/nwz031
- Karl, T. R., Knight, R. W., & Plummer, N. (1995). Trends in high-frequency climate variability in the twentieth century. *Nature*, 377(6546), 217–220. https://doi.org/10.1038/377217a0
- Kollas, C., Körner, C., & Randin, C. F. (2014). Spring frost and growing season length co-control the cold range limits of broad-leaved trees. *Journal of Biogeography*, *41*(4), 773–783. https://doi.org/10.1111/jbi.12238

Acknowledgments

This work was supported by The Second Tibetan Plateau Scientific Expedition and Research program (2019QZKK0102), the National Natural Science Foundation of China (42075019, 42075081), the Heavy Rain and Drought-Flood Disasters in Plateau and Basin Key Laboratory of Sichuan Province (SZKT201906). The authors thank CMIP6 community for contributing to the model outputs and China Meteorological Administration for the observational data. The authors also thank the anonymous reviewers and editor for their helpful comments.

- Körner, C., Basler, D., Hoch, G., Kollas, C., Lenz, A., Randin, C. F., et al. (2016). Where, why and how? Explaining the low-temperature range limits of temperate tree species. *Journal of Ecology*, 104(4), 1076–1088. https://doi.org/10.1111/1365-2745
- Lau, W., Kim, M., Kim, K., & Lee, W. (2010). Enhanced surface warming and accelerated snow melt in the Himalayas and Tibetan Plateau induced by absorbing aerosols. *Environmental Research Letters*, 5(2), 025204. https://doi.org/10.1088/1748-9326/5/2/025204
- Li, C., Bosch, C., Kang, S., Andersson, A., Chen, P., Zhang, Q., et al. (2016). Sources of black carbon to the Himalayan-Tibetan Plateau glaciers. *Nature Communications*, 7(1), 12574. https://doi.org/10.1038/ncomms12574
- Li, F., Wan, X., Wang, H., Orsolini, Y. J., Cong, Z., Gao, Y., & Kang, S. (2020). Arctic sea-ice loss intensifies aerosol transport to the Tibetan Plateau. Nature Climate Change, 10(11), 1037–1044. https://doi.org/10.1038/s41558-020-0881-2
- Li, Y., Huang, Y., Bergelson, J., Nordborg, M., & Borevitz, J. O. (2010). Association mapping of local climate-sensitive quantitative trait loci in Arabidopsis thaliana. Proceedings of the National Academy of Sciences of the United States of America, 107, 21199–21204. https://doi. org/10.1073/pnas.1007431107
- Liancourt, P., Song, X., Macek, M., Santrucek, J., & Dolezal, J. (2020). Plant's-eye view of temperature governs elevational distributions. *Global Change Biology*, 26(7), 4094–4103. https://doi.org/10.1111/gcb.15129
- Liu, X., Cheng, Z., Yan, L., & Yin, Z. Y. (2009). Elevation dependency of recent and future minimum surface air temperature trends in the Tibetan Plateau and its surroundings. *Global and Planetary Change*, 68(3), 164–174. https://doi.org/10.1016/j.gloplacha.2009.03.017
- Liu, X. D., & Chen, B. D. (2000). Climatic warming in the Tibetan Plateau during recent decades. International Journal of Climatology, 20(14), 1729–1742. https://doi.org/10.1002/1097-0088(20001130)20:14<1729</p>
- Mann, M. E., & Park, J. (1996). Greenhouse warming and changes in the seasonal cycle of temperature: Model versus observations. *Geophysical Research Letters*, 23, 1111–1114. https://doi.org/10.1029/96GL01066
- Marquis, B., Bergeron, Y., Simard, M., & Tremblay, F. (2020). Growing-season frost is a better predictor of tree growth than mean annual temperature in boreal mixedwood forest plantations. *Global Change Biology*, 00(11), 1–18. https://doi.org/10.1111/gcb.15327
- McKenzie Skiles, S., & Painter, T. H. (2018). Assessment of radiative forcing by light-absorbing particles in snow from in situ observations with radiative transfer modeling. *Journal of Hydrometeorology*, 19(8), 1397–1409. https://doi.org/10.1175/jhm-d-18-0072.1
- Minder, J. R., Letcher, T. W., & Liu, C. (2018). The character and causes of elevation-dependent warming in high-resolution simulations of rocky mountain climate change. *Journal of Climate*, 31, 2093–2113. https://doi.org/10.1175/JCLI-D-17-0321.1
- Paaijmans, K. P., Blanford, S., Bell, A. S., Blanford, J. I., Read, A. F., & Thomas, M. B. (2010). Influence of climate on malaria transmission depends on daily temperature variation. *Proceedings of the National Academy of Sciences of the United States of America*, 107(34), 15135– 15139. https://doi.org/10.1073/pnas.1006422107
- Paaijmans, K. P., Heinig, R. L., Seliga, R. A., Blanford, J. I., Blanford, S., Murdock, C. C., & Thomas, M. B. (2013). Temperature variation makes ectotherms more sensitive to climate change. *Global Change Biology*, 19(8), 2373–2380. https://doi.org/10.1111/gcb.12240
- Paik, S., & Min, S. K. (2020). Quantifying the anthropogenic greenhouse gas contribution to the observed spring snow-cover decline using the CMIP6 multimodel ensemble. *Journal of Climate*, 33(21), 9261–9269. https://doi.org/10.1175/jcli-d-20-0002.1
- Painter, T. H., Seidel, F. C., Bryant, A. C., McKenzie Skiles, S., & Rittger, K. (2013). Imaging spectroscopy of albedo and radiative forcing by light-absorbing impurities in mountain snow. *Journal of Geophysical Research: Atmospheres*, 118(17), 9511–9523. https://doi.org/10.1002/ jgrd.50520
- Palazzi, E., Filippi, L., & Hardenberg, J. v. (2017). Insights into elevation-dependent warming in the Tibetan Plateau-Himalayas from CMIP5 model simulations. *Climate Dynamics*, 48, 3991–4008. https://doi.org/10.1007/s00382-016-3316-z
- Pepin, N., Bradley, R. S., Diaz, H. F., Baraer, M., Caceres, E. B., Forsythe, N., et al. (2015). Elevation-dependent warming in mountain regions of the world. *Nature Climate Change*, 5, 424–430. https://doi.org/10.1038/nclimate2563
- Porter, W. P., Budaraju, S., Stewart, W. E., & Ramankutty, N. (2000). Calculating climate effects on birds and mammals: Impacts on biodiversity, conservation, population parameters, and global community structure. *American Zoologist*, 40(4), 597–630. https://doi.org/10.1093/ icb/40.4.597
- Qian, C., Fu, C., & Wu, Z. (2011). Changes in the amplitude of the temperature annual cycle in China and their implication for climate change research. Journal of Climate, 24(20), 5292–5302. https://doi.org/10.1175/JCLI-D-11-00006.1
- Qian, C., Wu, Z., Fu, C., & Wang, D. (2011). On changing El Niño: A view from time-varying annual cycle, interannual variability, and mean state. Journal of Climate, 24, 6486–6500. https://doi.org/10.1175/JCLI-D-10-05012.1
- Qian, C., & Zhang, X. (2015). Human influences on changes in the temperature seasonality in mid- to high-latitude land areas. *Journal of Climate*, 28, 5908–5921. https://doi.org/10.1175/JCLI-D-14-00821.1
- Qian, Y., Yasunari, T. J., Doherty, S. J., Flanner, M. G., Lau, W. K. M., Ming, J., et al. (2015). Light-absorbing particles in snow and ice: Measurement and modeling of climatic and hydrological impact. Advances in Atmospheric Sciences, 32(1), 64–91. https://doi.org/10.1007/ s00376-014-0010-0
- Qiu, J. (2008). China: The third pole. Nature, 454(7203), 393-396. https://doi.org/10.1038/454393a
- Rangwala, I., & Miller, J. R. (2012). Climate change in mountains: A review of elevation-dependent warming and its possible causes. *Climatic Change*, 114(3–4), 527–547. https://doi.org/10.1007/s10584-012-0419-3
- Rangwala, I., Miller, J. R., Russell, G. L., & Xu, M. (2010). Using a global climate model to evaluate the influences of water vapor, snow cover and atmospheric aerosol on warming in the Tibetan Plateau during the twenty-first century. *Climate Dynamics*, 34(6), 859–872. https://doi. org/10.1007/s00382-009-0564-1
- Rangwala, I., Miller, J. R., & Xu, M. (2009). Warming in the Tibetan plateau: Possible influences of the changes in surface water vapor. Geophysical Research Letters, 36, L06703. https://doi.org/10.1029/2009GL037245
- Rangwala, I., Sinsky, E., & Miller, R. J. (2013). Amplified warming projections for high altitude regions of the Northern hemisphere mid-latitudes from CMIP5 models. *Environmental Research Letters*, 8, 024040. https://doi.org/10.1088/1748-9326/8/2/024040
- Ruel, J. J., & Ayres, M. P. (1999). Jensen's inequality predicts effects of environmental variation. *Trends in Ecology & Evolution*, 14(9), 361–366. https://doi.org/10.1016/s0169-5347(99)01664-x
- Santer, B. D., Mikolajewicz, U., Brüggemann, W., Cubasch, U., Hasselmann, K., Höck, H., et al. (1995). Ocean variability and its influence on the detectability of greenhouse warming signals. *Journal of Geophysical Research*, 100, 10693–10725. https://doi.org/10.1029/95JC00683
- Santer, B. D., Po-Chedley, S., Zelinka, M. D., Cvijanovic, I., Bonfils, C., Durack, P. J., et al. (2018). Human influence on the seasonal cycle of tropospheric temperature. *Science*, 361(6399), eaas8806. https://doi.org/10.1126/science.aas8806
- Sarangi, C., Qian, Y., Rittger, K., Ruby Leung, L., Chand, D., Bormann, K. J., & Painter, T. H. (2020). Dust dominates high-altitude snow darkening and melt over high-mountain Asia. *Nature Climate Change*, 10(11), 1045–1051. https://doi.org/10.1038/s41558-020-00909-3
- Seddon, A. W. R., Macias-Fauria, M., Long, P. R., Benz, D., & Willis, K. J. (2016). Sensitivity of global terrestrial ecosystems to climate variability. *Nature*, 531(7593), 229–232. https://doi.org/10.1038/nature16986

- Seong, M. G., Min, S. K., Kim, Y. H., Zhang, X., & Sun, Y. (2021). Anthropogenic greenhouse gas and aerosol contributions to extreme temperature changes during 1951–2015. *Journal of Climate*, 34(3), 857–870. https://doi.org/10.1175/jcli-d-19-1023.1
- Shrestha, N., Su, X., Xu, X., & Wang, Z. (2018). The drivers of high Rhododendron diversity in south-west China: Does seasonality matter? Journal of Biogeography, 45(2), 438–447. https://doi.org/10.1111/jbi.13136
- Silva de Miranda, P. L., Oliveira-Filho, A. T., Pennington, R. T., Neves, D. M., Baker, T. R., & Dexter, K. G. (2018). Using tree species inventories to map biomes and assess their climatic overlaps in lowland tropical South America. *Global Ecology and Biogeography*, 27(8), 899–912. https://doi.org/10.1111/geb.12749
- Stine, A. R., & Huybers, P. (2012). Changes in the seasonal cycle of temperature and atmospheric circulation. Journal of Climate, 25(21), 7362–7380. https://doi.org/10.1175/JCLI-D-11-00470.1

Thakuri, S., Dahal, S., Shrestha, D., Guyennon, N., Romano, E., Colombo, N., & Salerno, F. (2019). Elevation-dependent warming of maximum air temperature in Nepal during 1976–2015. *Atmospheric Research*, 228, 261–269. https://doi.org/10.1016/j.atmosres.2019.06.006

Thompson, R. (1995). Complex demodulation and the estimation of the changing continentality of Europe's climate. International Journal of Climatology, 15(2), 175–185. https://doi.org/10.1002/joc.3370150204

Thomson, D. J. (1995). The seasons, global temperature and precession. *Science*, 268(5207), 59–68. https://doi.org/10.1126/science.268.5207.59 Thomson, D. J. (2009). Shifts in season. *Nature*, 457(7228), 391–392. https://doi.org/10.1038/457391a

Vasseur, D. A., DeLong, J. P., Gilbert, B., Greig, H. S., Harley, C. D. G., McCann, K. S., et al. (2014). Increased temperature variation poses a greater risk to species than climate warming. *Proceedings of the Royal Society B*, 281(1779), 20132612. https://doi.org/10.1098/rspb.2013.2612

Vitra, A., Lenz, A., & Vitasse, Y. (2017). Frost hardening and dehardening potential in temperate trees from winter to budburst. New Phytologist, 216(1), 113–123. https://doi.org/10.1111/nph.14698

Wallace, C. J., & Osborn, T. J. (2002). Recent and future modulation of the annual cycle. Climate Research, 22, 1–11. https://doi.org/10.3354/ cr022001

Wan, H., Zhang, X., Zwiers, F., & Min, S.-K. (2015). Attributing northern high-latitude precipitation change over the period 1966–2005 to human influence. *Climate Dynamics*, 45, 1713–1726. https://doi.org/10.1007/s00382-014-2423-y

Wan, X., Kang, S., Li, Q., Rupakheti, D., Zhang, Q., Guo, J., et al. (2017). Organic molecular tracers in the atmospheric aerosols from Lumbini, Nepal, in the northern Indo-Gangetic Plain: Influence of biomass burning. Atmospheric Chemistry and Physics, 17(14), 8867–8885. https:// doi.org/10.5194/acp-17-8867-2017

Wang, G., & Dillon, M. E. (2014). Recent geographic convergence in diurnal and annual temperature cycling flattens global thermal profiles. *Nature Climate Change*, 4(11), 988–992. https://doi.org/10.1038/NCLIMATE2378

Wen, Z., Wu, Y., Cheng, J., Cai, T., Du, Y., Ge, D., et al. (2018). Abundance of small mammals correlates with their elevational range sizes and elevational distributions in the subtropics. *Ecography*, 41(11), 1888–1898. https://doi.org/10.1111/ecog.03558

Xia, X., Zong, X., Cong, Z., Chen, H., Kang, S., & Wang, P. (2011). Baseline continental aerosol over the central Tibetan plateau and a case study of aerosol transport from South Asia. Atmospheric Environment, 45(39), 7370–7378. https://doi.org/10.1016/j.atmosenv.2011.07.067

Xu, B. Q., Cao, J. J., Hansen, J., Yao, T. D., Joswia, D. R., Wang, N. L., et al. (2009). Black soot and the survival of Tibetan glaciers. Proceedings of the National Academy of Sciences of the United States of America, 106(52), 22114–22118. https://doi.org/10.1073/pnas.0910444106

Yao, T., Xue, Y., Chen, D., Chen, F., Thompson, L., Cui, P., et al. (2019). Recent third pole's rapid warming accompanies cryospheric melt and water cycle intensification and interactions between monsoon and environment: Multidisciplinary approach with observations, modeling, and analysis. *Bulletin of the American Meteorological Society*, 100(3), 423–444. https://doi.org/10.1175/bams-d-17-0057.1

Yao, T. D., Thompson, L., Mosbrugger, V., Zhang, F., Ma, Y., Luo, T., et al. (2012). Third Pole environment (TPE). *Environmental Development*, 3, 52–64. https://doi.org/10.1016/j.envdev.2012.04.002

Yao, T. D., Thompson, L., Yang, W., Yu, W. S., Gao, Y., Guo, X. J., et al. (2012). Different glacier status with atmospheric circulations in Tibetan Plateau and surroundings. *Nature Climate Change*, 2(9), 663–667. https://doi.org/10.1038/nclimate1580

You, Q. L., Chen, D. L., Wu, F. Y., Pepin, N., Cai, Z. Y., Ahrens, B., et al. (2020). Elevation dependent warming over the Tibetan Plateau: Patterns, mechanisms and perspectives. *Earth-Science Reviews*, 210, 103349. https://doi.org/10.1016/j.earscirev.2020.103349

Stine, A. R., Huybers, P., & Fung, I. Y. (2009). Changes in the phase of the annual cycle of surface temperature. Nature, 457(7228), 435–440. https://doi.org/10.1038/nature07675